Author’s response to reviews

Title: PRESERVATION OF RENAL FUNCTION IN CARDIAC SURGERY PATIENTS WITH LOW CARDIAC OUTPUT SYNDROME: LEVOSIMENDAN VS BETA AGONISTS

Authors:

JOSE LUIS GUERRERO ORRIACH (guerreroorriach@gmail.com)
ISABEL Arce (isanavarroarce@gmail.com)
PAULA RODRIGUEZ (paulahernandez81@gmail.com)
aida Ponferrada (aidaraigonp@gmail.com)
ALFREDO Manso (alfredomalomanso@gmail.com)
MARTA ALIAGA (martaram85@gmail.com)
ALICIA FERNANDEZ (aliramfdez@gmail.com)
JUAN JOSE BELMONTE (jjescalonabelmonte@gmail.com)
INMACULADA ESTEVEZ (inmabellido@gmail.com)
AURELIO LUQUE (jagomez@uma.es)
RAFAEL SERRANO (rafbaser@gmail.com)
CARMEN MEDINA (carmentm_86@hotmail.com)
MANUEL navarro (mrubionav@gmail.com)
jose mañas (jose.cruz.sspa@juntadeandalucia.es)

Version: 1 Date: 29 Oct 2018

Author’s response to reviews:

I send again our article wit all the changes proposed.

First one thank you very much for all your comments; and the first answer to the first question is that we wrote the type of study wrong; it is a prospective study, not randomized. We scheduled the patients if they were in LOCS, after this the distribution of cases with beta agonist treatment and levosimendan was symilar.
Thanks again.